Author's response to reviews

Title: Maternal fish and shellfish intake and pregnancy outcomes. A prospective cohort study in Brittany, France.

Authors:

Laurence Guldner (laurence.guldner@univ-rennes1.fr)
Christine Monfort (christine.monfort@rennes.inserm.fr)
Florence Rouget (florence.rouget@univ-rennes1.fr)
Ronan Garlantezec (ronan.garlantezec@univ-rennes1.fr)
Sylvaine Cordier (sylvaine.cordier@rennes.inserm.fr)

Version: 4 Date: 18 October 2007

Author's response to reviews: see over
Ref: Manuscript “Maternal fish and shellfish intake and pregnancy outcomes. A prospective cohort study in Brittany, France.”

Sir,

Please find the revised version of the manuscript referenced above.

The sentence p.5 line 7-8 was rephrased as requested.

10-4 was changed in $10^{-4}$ page 6 line 24 and also in table 1 (p.15 line 12).

We replaced “n” by “N” and “P” by “p” in table 1.

Capitals where removed for all section headings p.2, 3, 4, 6, 7, 10 and 11.

Following a comment of one co-author, we also changed the term “infants” in “newborns” p.5 line 9, p.6 line 27, p.7 line 4 and p.8 line 2, as examinations concerns babies under the age of 29 days.

Yours sincerely,

Laurence Guldner